Author’s response to reviews

Title: Participatory Science and Innovation for Improved Sanitation and Hygiene: Process and Outcome Evaluation of Project SHINE, a School-based Intervention in Rural Tanzania.

Authors:

Erin Hetherington (elhether@ucalgary.ca)
Matthijs Eggers (m.eggers@maastrichtuniversity.nl)
Joyce Wamoyi (Jwamoyi@hotmail.com)
Jennifer Hatfield (jhatfiel@ucalgary.ca)
Mange Manyama (mmajfe@hotmail.com)
Susan Kutz (skutz@ucalgary.ca)
Sheri Bastien (sheri.lee.bastien@nmbu.no)

Version: 1 Date: 23 Sep 2016

Author’s response to reviews:

Dear Editors,

As requested we have completed the following revisions to our submission:

1. The title page and abstract have been added to the manuscript.
2. Certificates of ethical clearance for the study have been removed.

Please do not hesitate to get in touch if you have any further questions.

Kind regards,

Sheri Bastien, PhD